Author's response to reviews

Title: Fascin overexpression promotes neoplastic progression in Oral Squamous Cell Carcinoma

Authors:

Hunain Alam (hunainalam@gmail.com)
Amruta V Bhate (amrutabhate@gmail.com)
Prakash Gangadaran (prakashonmail@gmail.com)
Sharda S Sawant (ssswant@actrec.gov.in)
Shimul Salot (ssshimul10@yahoo.com)
Lalit Sehgal (lalit.genetics@gmail.com)
Prerana P Dange (dange.prerana@gmail.com)
Devendra A Chaukar (dchaukar@gmail.com)
Anil K D&$@cruz (docdrcruz@gmail.com)
Sadhna Kannan (skannant@actrec.gov.in)
Rajiv Gude (rgude@actrec.gov.in)
Shubhada Kane (drsvkane@rediffmail.com)
Sorab N Dalal (sdalal@actrec.gov.in)
Milind M Vaidya (mvaidya@actrec.gov.in)

Version: 5 Date: 2 January 2012

Author's response to reviews: see over
To,
The Associate Editor,
BMC Cancer,

Dear Prof. Roskelley,

Thank you for sending us the review of our manuscript “Fascin overexpression promotes neoplastic progression in OSCC”. We have addressed all the comments of the reviewers. Point-by-point rebuttal to these comments has been uploaded as separate file. As you have suggested, we have answered the concerns raised by reviewer 1 regarding microscopy. We have changed the figure wherever we found it was necessary. We have also addressed the issue regarding reproducibility raised by reviewer 2 and modified the text related to EMT.

We have restructured the abstract according to editorial requirement. We have also included a separate CONCLUSIONS SECTION as the last section of the text.

I hope modifications made by us meet the requirements for the publication in BMC Cancer.

Thanking you,

Sincerely,

Milind M. Vaidya Ph.D.
KS115, ACTREC, CRI, Tata Memorial Centre
Kharghar Node, Navi Mumbai 410210.
Maharashtra, India
Email: mvaidy@actrec.gov.in
Phone #: 91-22-27405055